Author’s response to reviews

Title: Tibiototalocalcaneal (TTC) arthrodesis with reverse PHILOS plate and medial cannulated screws with lateral approach

Authors:

Jian Fan (qidongfanjian@sina.com)

Xing Zhang (2439299822@qq.com)

Yong Luo (fanjian7406@sohu.com)

Gui-wu You (youguiwu123@163.com)

NG WAI KING (weiqing@sina.com)

Yun-feng Yang (qidongfanyihang@126.com)

Version: 2 Date: 15 Mar 2017

Author’s response to reviews:

BMSD-D-16-00797R1

Tibiototalocalcaneal (TTC) arthrodesis with reverse PHILOS plate and medial cannulated screws with lateral approach

Jian Fan; Xing Zhang; Yong Luo; Gui-wu You; NG WAI KING; Yun-feng Yang

BMC Musculoskeletal Disorders

Dear Dr Yang,

Your manuscript "Tibiototalocalcaneal (TTC) arthrodesis with reverse PHILOS plate and medial cannulated screws with lateral approach" (BMSD-D-16-00797R1) has been assessed by our reviewers. Based on these reports, and my own assessment as Editor, I am pleased to inform you that it is potentially acceptable for publication in BMC Musculoskeletal Disorders, once you have carried out some essential revisions suggested:
Please make the following revisions to your manuscript:

1. Due to the list of authors for your manuscript changing, we would appreciate it if you could please submit a change of authorship form with your revision.

For full instructions on how to complete this form, please see:

http://www.biomedcentral.com/getpublished/editorial-policies#authorship

Answer: We have completed the change of form of authorship.

2. Please add a 'Declarations' heading to your manuscript before the references section. Please ensure that this section contains all of the following sub-sections:

- Ethics and consent to participate

Please include a statement on ethics approval and consent. Please clarify if this was a novel surgical method or an analysis of new technique.

Please include the name of the ethics committee that approved the study and the committee’s reference number if appropriate. If you did not need formal ethics approval please confirm that this complies with national guidelines and provide a reference which supports this. Alternatively, supply a statement that says that a local ethics committee ruled that no formal ethics approval was required in this particular case.

If consent was not required, please provide details of why this was wavered and the institution/committee who approved this.
- Consent to publish

If your manuscript contains any individual person’s data in any form (including individual details, images or videos), consent to publish must be obtained from that person, or in the case of children, their parent or legal guardian. Please provide details of this consent in this section.

- Competing interests

Please list any competing interests of the authors

- Funding  (already provided)

- Authors' contributions

We use this information to note which author contributed what to the study. An ‘author’ is generally considered to be someone who has made substantive intellectual contributions to a published study. We respectfully request you please amend this statement to adhere to the guidelines for authorship that are applicable in your research field or, in the absence of any guidelines, to the International Committee of Medical Journal Editors (ICMJE) guidelines. According to the ICMJE guidelines, to qualify as an author one should have:

- made substantial contributions to conception and design, or acquisition of data, or analysis and interpretation of data;

- been involved in drafting the manuscript or revising it critically for important intellectual content;

- given final approval of the version to be published. Each author should have participated sufficiently in the work to take public responsibility for appropriate portions of the content; and
- agreed to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved.

Acquisition of funding, collection of data, or general supervision of the research group, alone, does not usually justify authorship.

Please include a statement in Authors’ contributions clearly indicating that all authors read and approved the final version of the manuscript.

- Availability of data and materials

Please detail where the raw data supporting your findings can be found (including information found in any supplementary files). If you do not wish to share your data, please state that data will not be shared, and state the reason. If all the data supporting your findings is contained within the manuscript, please state so. Please note that identifying/confidential patient data should not be shared.

Please note that we strongly encourage authors to share their data in a publicly available repository as this increases the visibility of your data/study. For more information, please see our editorial policies: http://www.biomedcentral.com/getpublished/editorial-policies#availability+of+data+and+materials

- Acknowledgements

Please only use this section to acknowledge anyone who contributed towards the article who does not meet the criteria for authorship including anyone who provided professional writing services or materials.

Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements section. If no acknowledgements are necessary then please state 'None'.
- Authors' information  (optional)

You may choose to use this section to include any relevant information about the author(s) that may aid the reader's interpretation of the article, and understand the standpoint of the author(s). This may include details about the authors' qualifications, current positions they hold at institutions or societies, or any other relevant background information. Please refer to authors using their initials. Note this section should not be used to describe any competing interests.

Answer: We have revised and add these parts according with your suggestion and margined with red word in the revised paper.